Author's response to reviews

Title: RNA-binding protein RNPC1: acting as a tumor suppressor in breast cancer

Authors:

Jin-Qiu Xue (xuejinqiu_1986@163.com)
Tian-Song Xia (xiatsswms@163.com)
Xiu-Qing Liang (517644520@qq.com)
Wenbin Zhou (zhouwenbin_1984@foxmail.com)
Lin Cheng (694427729@qq.com)
Liang Shi (337039607@qq.com)
Ying Wang (602184984@qq.com)
Qiang Ding (dingqiang@njmu.edu.cn)

Version: 6 Date: 19 April 2014

Author's response to reviews: see over
Cover letter

Dear Editors:

Thank you for the editor’s contribution and comments concerning to our manuscript entitled "RNA-binding protein RNPC1: acting as a tumor suppressor in breast cancer" (ID: 6460663561151965). Those comments are valuable and very helpful for improving our paper, as well as our further study. We have addressed editor’s comment carefully and made correction which we hope could meet the approval.

We appreciate Editor’s warm work earnestly, and revised my paper point-by-point.

Once again, thank you and the reviewers very much for your help.

Best Regards,

Qiang Ding

Jiangsu Breast Disease Center, The First Affiliated Hospital with Nanjing Medical University, 300 Guangzhou Road, 210029 Nanjing, China.

Email: dingqiang@njmu.edu.cn
Responds to the editor’s comments

All the corrections and the responds to the editor’s comments was displayed point-by-point as follow:

Response to editor (the changes according to editor’s comments is marked in blue in the Revised Manuscript with Tracked Changes)

Q1: Please could you clarify whether written informed consent was obtained from patients prior to tissue collection.
Response: We are appreciated the editor’s suggestion to the manuscript. As the editor’s mentioned, we clarified that written informed consent was obtained from patients prior to tissue collection. (Page5- 6, Line 135-138)

Q2: Please could you format the dates "2006.2 to 2009.8" correctly - line 126.
Response: We are appreciated the editor’s correction. We have revised the date format "2006.2 to 2009.8" correctly - line 126. (Page5, Line 126)